Specific Comments to Authors: The authors reported a case with a title "Hepatic Cavernous Hemangioma: A Potential Pitfall on 18F-FDG PET/CT". The manuscript is well-written with adequate reference to the current medical literature in the field. The content of the text is satisfactory enough and the figures are informative.

Answer: We appreciate the positive comments from the reviewer.